Author's response to reviews

Title: Pro- and antisaccades in children elicited by visual and acoustic targets - does modality matter?

Authors:

   Johanna Goepel (hanna@wolkenkind.de)
   Stefanie C Biehl (Biehl_S@klinik.uni-wuerzburg.de)
   Johanna Kissler (Johanna.Kissler@uni-konstanz.de)
   Isabella Paul-Jordanov (Isabella.Paul-Jordanov@uni-konstanz.de)

Version: 5 Date: 13 December 2011

Author's response to reviews: see over
Dear Editorial Production Team,

I confirm that all authors read and approved the final manuscript in the author Contributions section. We included the country details in all affiliations and removed all forms of placement text; so that I hope that all formatting changes were made. Thank you for accepting our manuscript for publication in the BMC Pediatrics.

Sincerely yours,

Johanna Goepel